Author's response to reviews

Title: Targeting mTOR/p70S6K/glycolysis signaling pathway restores glucocorticoid sensitivity to 4E-BP1 null Burkitt Lymphoma

Authors:

Ling Gu (guling00@163.com)
Liping Xie (lipinxie@yeah.net)
Chuan Zuo (zcsiren@163.com)
Zhigui Ma (zhiguima66@163.com)
Yanle Zhang (yanlizhang@yeah.net)
Yipeng Zhu (yipingzhu18@126.com)
Gao Ju (gaoju651220@163.com)

Version: 6 Date: 12 June 2015

Author's response to reviews: see over
Dear Editor,

Thanks very much for your kind work and consideration on publication of our manuscript. On behalf of my co-authors, we would like to express our great appreciation to you and reviewers.

We have studied the comments and suggestions carefully and have made revisions which were highlighted in yellow in the paper.

We carefully checked and revised our formatting against the guidelines.

Thank you and best regards.

Sincerely,

Ling Gu, MD, PhD
Laboratory of Hematology/Oncology
West China Second University Hospital
Sichuan University
Chengdu, 610041, China
Response to Dr. Melba Jaramillo

Thank you very much for your positive and constructive suggestions.

The manuscript has been carefully read through and all the grammar errors we could find in the text have been corrected.